Author’s response to reviews

Title: Antihistamine Effects and Safety of Fexofenadine: A Systematic Review and Meta-analysis of Randomized Controlled Trials

Authors:
Cheng-zhi Huang (15626044356@163.com)
Zhi-hui Jiang (jandsphy@163.com)
Jian Wang (13889909318@139.com)
Yue Luo (15622720399@163.com)
Hua Peng (doctorpeng2006@hotmail.com)

Version: 1 Date: 14 Feb 2019

Author’s response to reviews:

Dear Editor,

We would like to thank you for giving us a chance to resubmit the paper and giving us constructive suggestions which would help us in depth to improve the quality of the paper. Here we submit a new version of our manuscript with the title “Antihistamine Effects and Safety of Fexofenadine: A Systematic Review and Meta-analysis of Randomized Controlled Trials”, which has been modified according to your suggestions. Efforts were also made to correct the mistakes.

Sincerely yours,

Hua Peng